Author's response to reviews

Title: Influencing factors on serum 25-hydroxyvitamin D3 levels in Japanese chronic hepatitis C patients

Authors:

Masanori Atsukawa (momogachi@yahoo.co.jp)
Akihito Tsubota (atsubo@iikei.ac.jp)
Noritomo Shimada (noritomos@hotmail.com)
Kai Yoshizawa (kaiyoshiza@yahoo.co.jp)
Hiroshi Abe (hirosihiabe222@aol.com)
Toru Asano (tasanobkttokyo@yahoo.co.jp)
Yusuke Ohkubo (yusuke-okubou@mtg.biglobe.ne.jp)
Masahiro Araki (ma-araki@chubyoin.pref.ibaraki.jp)
Tadashi Ikekami (iikegami@tokyo-med.ac.jp)
Chisa Kondo (s8042@hotmail.co.jp)
Norio Itokawa (mid@mtc.biglobe.ne.jp)
Ai Nakagawa (a0520i@nms.ac.jp)
Taeang Arai (taeang@yahoo.co.jp)
Yoko Matsushita (yokoyymym18s@yahoo.co.jp)
Katsuhisa Nakatsuka (kaynet@nms.ac.jp)
Tomomi Furhata (tomomif@faculty.chiba-u.jp)
Yoshimichi Chuganji (chuganji@gmail.com)
Yasushi Matsuzaki (ymatsuzaki-gi@umin.ac.jp)
Yoshio Aizawa (aichanyoshi@yahoo.co.jp)
Katsuhiko Iwakiri (k-iwa@nms.ac.jp)

Version: 4 Date: 12 July 2015

Author's response to reviews:

Point-by-point
We are grateful to the editor for the helpful comments. We have revised the manuscript accordingly, and hope that it is now suitable for publication.

Editor’s comment
The Associate Editor has now recommended your manuscript for acceptance. Before we can accept your paper, we need you to do some minor editorial revisions:

1) Please name all of the authors in the Author contribution statement
   #As suggested by the editor, we rewrote ‘author’s contribution’.

2) Please remove the Figure titles from the Figures. Figures should just be the images themselves.
   #As suggested by the editor, we removed figure’s titles from figures.
3) Please give an individual email address for Tomomi Furihata and ensure that his email is corrected in the submission system.
#We corrected Tomomi Furihata’s e-mail address.